Author’s response to reviews

Title: Comparing two common surgical approaches to the hip for hemiarthroplasty. A scoping review

Authors:

James Fullam (j.a.fullam@exeter.ac.uk)
Paraskevas Theodosi (paraskevas.theodosi@nhs.net)
John Charity (john.charity@nhs.net)
Victoria Goodwin (V.Goodwin@exeter.ac.uk)

Version: 2 Date: 21 Feb 2019

Author’s response to reviews:

RE: BSUR-D-18-00425

Comparing two common surgical approaches to the hip for hemiarthroplasty. A scoping review
James Fullam; Paraskevas G Theodosi; John Charity; Victoria Goodwin BMC Surgery

21/12/2018

Dear Mr O’Cueller,

We thank the reviewers for their comments and note that all appear satisfied with changes and clarifications with respect to their initial review. Please see below specific responses (in red) to each of the points raised by the editor:

Editor Comments:

1. Figure legends
Please provide figure legends under the separate heading, 'Figure Legends,' which should be placed after the References. If Figure titles/legends are within the main text or the figure files, please move them. Figure files should contain only the image, as well as any associated keys/annotations.

The title for Figure 1 has been removed from the main text and can now be located under the references.

2. Split funding/acknowledgements statements

Please split the Funding/Acknowledgement statements into two separate statements. If you have no acknowledgements please put ‘Not Applicable’ in this section.

Funding/Acknowledgements has been split

3. Authors’ contributions – missing authors

The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found in the online journal submission guidelines.

Individual author contributions have now been specified.

4. Consent for publication

We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Included section and stated “Not applicable”
5. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Completed

Kind regards

James Fullam